Author's response to reviews

Title: The psychometric properties of three self-report screening instruments for identifying frail older people in the community

Authors:

Silke F Metzelthin (s.metzelthin@zw.unimaas.nl)
Ramon Daniëls (r.daniels@hszuyd.nl)
Erik van Rossum (e.van.rossum@hszuyd.nl)
Luc de Witte (l.de.witte@hszuyd.nl)
Wim JA van den Heuvel (heuvelwim@hotmail.com)
Gertrudis IJM Kempen (g.kempen@zw.unimaas.nl)

Version: 9 Date: 22 March 2010

Author's response to reviews: see over
22nd March, 2010

Dear Dr. Kerr,

We have made the following final change to our manuscript:

1) We have changed reference 17. We refer to the article of Gobbens and colleagues in the following way:


We hope that the change fulfils your expectations.

Sincerely yours,
on behalf of the co-authors,

Silke F. Metzelthin